Author's response to reviews

Title: Oxidative stress level is not associated with survival in terminally ill cancer patients: a preliminary study

Authors:

Chang Hwan Yeom (lymphych@hanmail.net)
Youn Seon Choi (younseon@korea.ac.kr)
Hong Yup Ahn (ahn@dongquk.edu)
Su Hey Lee (ilohyang@hanmail.net)
In Cheol Hwang (spfe0211@gmail.com)

Version: 6 Date: 11 March 2014

Author's response to reviews: see over
BMC palliative care Editorial Office

Dear Editor:

It is a great pleasure for us to receive an e-mail of acceptance. Following your requests, we added the names of ethics committee and the approval numbers in Page 3.

Sincerely yours,

In Cheol Hwang, MD, MPH, PhD
Palliative Care Unit, Incheon Regional Cancer Center
E-mail address: spfe0211@gmail.com